

G OPEN ACCESS

Citation: Suri S, Chiesa ST, Zsoldos E, Mackay CE, Filippini N, Griffanti L, et al. (2020) Associations between arterial stiffening and brain structure, perfusion, and cognition in the Whitehall II Imaging Sub-study: A retrospective cohort study. PLoS Med 17(12): e1003467. https://doi.org/10.1371/ journal.pmed.1003467

Academic Editor: Joshua Z. Willey, Columbia University, UNITED STATES

Received: July 1, 2020

Accepted: November 13, 2020

Published: December 29, 2020

Copyright: © 2020 Suri et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The study follows Medical Research Council data sharing policies (https://mrc.ukri.org/research/policies-andguidance-for-researchers/data-sharing/). In accordance with these guidelines, data from the Whitehall II Study and the Imaging Sub-study are accessible via a formal application on the Dementias Platform UK portal (https://portal. dementiasplatform.uk/). RESEARCH ARTICLE

Associations between arterial stiffening and brain structure, perfusion, and cognition in the Whitehall II Imaging Sub-study: A retrospective cohort study

Sana Suri^{[0],2‡}*, Scott T. Chiesa^{3‡}, Enikő Zsoldos^{1,2}, Clare E. Mackay^{1,2}, Nicola Filippini^{1,2}, Ludovica Griffanti^{1,2}, Abda Mahmood¹, Archana Singh-Manoux^{4,5}, Martin J. Shipley⁴, Eric J. Brunner⁴, Mika Kivimäki⁴, John E. Deanfield^{3‡}, Klaus P. Ebmeier^{1‡}

 Department of Psychiatry, Warneford Hospital, University of Oxford, Oxford, United Kingdom, 2 Oxford Centre for Human Brain Activity, Wellcome Centre for Integrative Neuroimaging, University of Oxford, Oxford, United Kingdom, 3 Institute of Cardiovascular Science, University College London, London, United Kingdom,
Department of Epidemiology and Public Health, University College London, London, United Kingdom,
Inserm U1153, Epidemiology of Ageing and Neurodegenerative diseases, Université de Paris, Paris, France

\$\\$S and STC share first authorship on this work. JED and KPE share last authorship on this work. \$\\$sana.suri@psych.ox.ac.uk

Abstract

Background

Aortic stiffness is closely linked with cardiovascular diseases (CVDs), but recent studies suggest that it is also a risk factor for cognitive decline and dementia. However, the brain changes underlying this risk are unclear. We examined whether aortic stiffening during a 4-year follow-up in mid-to-late life was associated with brain structure and cognition in the Whitehall II Imaging Sub-study.

Methods and findings

The Whitehall II Imaging cohort is a randomly selected subset of the ongoing Whitehall II Study, for which participants have received clinical follow-ups for 30 years, across 12 phases. Aortic pulse wave velocity (PWV) was measured in 2007–2009 (Phase 9) and at a 4-year follow-up in 2012–2013 (Phase 11). Between 2012 and 2016 (Imaging Phase), participants received a multimodal 3T brain magnetic resonance imaging (MRI) scan and cognitive tests. Participants were selected if they had no clinical diagnosis of dementia and no gross brain structural abnormalities. Voxel-based analyses were used to assess grey matter (GM) volume, white matter (WM) microstructure (fractional anisotropy (FA) and diffusivity), white matter lesions (WMLs), and cerebral blood flow (CBF). Cognitive outcomes were performance on verbal memory, semantic fluency, working memory, and executive function tests. Of 542 participants, 444 (81.9%) were men. The mean (SD) age was 63.9 (5.2) years at the baseline Phase 9 examination, 68.0 (5.2) at Phase 11, and 69.8 (5.2) at the Imaging Phase. Voxel-based analysis revealed that faster rates of aortic stiffening in mid-to-late life

Funding: The Whitehall II study is supported by the British Heart Foundation (RG/16/11/32334), UK Medical Research Council (K013351) and US National Institute on Aging (R01AG013196; R01AG034454; R01AG056477). The Whitehall II Imaging Sub-study was supported by the UK Medical Research Council (MRC) grants "Predicting MRI abnormalities with longitudinal data of the Whitehall II Sub-study" (G1001354; PI KPE; ClinicalTrials.gov Identifier: NCT03335696), and "Adult Determinants of Late Life Depression, Cognitive Decline and Physical Functioning - The Whitehall II Ageing Study" (MR/K013351/1; PI: MK). Work on this study was supported by European Commission (Horizon 2020 grant "Lifebrain", 732592; co-PI KPE), the HDH Wills 1965 Charitable Trust (1117747; PI KPE) and the UK National Institute of Health Research (NIHR) Oxford Health Biomedical Research Centre (BRC). The Wellcome Centre for Integrative Neuroimaging (WIN) is supported by core funding from the Wellcome Trust (203139/Z/16/Z). The authors report the following funding: SS (Alzheimer's Society Research Fellowship (Grant Number 441), KPE and EZ (UK Medical Research Council (G1001354, MR/K013351/), HDH Wills 1965 Charitable Trust (1117747), and the EU Horizon 2020 Grant: Healthy minds 0-100 years: Optimising the use of European brain imaging cohorts (Lifebrain), Grant Agreement No. 732592), MK (UK MRC (MR/K013351/1, MR/R024227/1, MR/S011676/1), National Institute on Aging (NIH), US (R01AG056477), NordForsk (75021), Academy of Finland (311492), Helsinki Institute of Life Science Fellowship (H970)), LG (Monument Trust Discovery Award from Parkinson's UK (J-1403) and the MRC Dementias Platform UK (MR/ L023784/2)), ASM (NIH (R01AG056477, R01AG062553). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: JED reports provision of medical consulting for the Brain Protection Company Ltd. The remaining authors have declared that no competing interests exist.

Abbreviations: AD, axial diffusivity; APOE4, apolipoprotein E4; BMI, body mass index; CBF, cerebral blood flow; CVD, cardiovascular disease; DTI, diffusion tensor imaging; FA, fractional anisotropy; FLAIR, fluid-attenuated inversion recovery; FSL, FMRIB Software Library; FWE, family-wise error; GLM, general linear model; GM, grey matter; HVLT-R, Hopkins Verbal Learning Test–Revised; MAP, mean arterial pressure; MD, were associated with poor WM microstructure, viz. lower FA, higher mean, and radial diffusivity (RD) in 23.9%, 11.8%, and 22.2% of WM tracts, respectively, including the corpus callosum, corona radiata, superior longitudinal fasciculus, and corticospinal tracts. Similar voxel-wise associations were also observed with follow-up aortic stiffness. Moreover, lower mean global FA was associated with faster rates of aortic stiffening (B = -5.65, 95% CI -9.75, -1.54, Bonferroni-corrected p < 0.0125) and higher follow-up aortic stiffness (B = -1.12, 95% CI -1.95, -0.29, Bonferroni-corrected p < 0.0125). In a subset of 112 participants who received arterial spin labelling scans, faster aortic stiffening was also related to lower cerebral perfusion in 18.4% of GM, with associations surviving Bonferroni corrections in the frontal (B = -10.85, 95% CI -17.91, -3.79, p < 0.0125) and parietal lobes (B = -12.75, 95% CI -21.58, -3.91, p < 0.0125). No associations with GM volume or WMLs were observed. Further, higher baseline aortic stiffness was associated with poor semantic fluency (B = -0.47, 95% CI -0.76 to -0.18, Bonferroni-corrected p < 0.007) and verbal learning outcomes (B = -0.36, 95% CI -0.60 to -0.12, Bonferroni-corrected p < 0.007). As with all observational studies, it was not possible to infer causal associations. The generalisability of the findings may be limited by the gender imbalance, high educational attainment, survival bias, and lack of ethnic and socioeconomic diversity in this cohort.

Conclusions

Our findings indicate that faster rates of aortic stiffening in mid-to-late life were associated with poor brain WM microstructural integrity and reduced cerebral perfusion, likely due to increased transmission of pulsatile energy to the delicate cerebral microvasculature. Strategies to prevent arterial stiffening prior to this point may be required to offer cognitive benefit in older age.

Trial registration

ClinicalTrials.gov NCT03335696

Author summary

Why was this study done?

- Stiffening of large arteries such as the aorta has been closely linked to the onset of heart diseases, but recent studies show that this may also increase risk for dementia.
- Understanding the brain changes which underlie this risk could help identify strategies to prevent or delay dementia in old age.

What did the researchers do and find?

• We investigated 542 community-dwelling older adults who received 2 measurements of aortic stiffness, first at approximately 64 years old and again at approximately 68 years old.

mean diffusivity; MoCA, Montreal Cognitive Assessment; MRI, magnetic resonance imaging; pCASL, pseudocontinuous arterial spin labelling; PWV, pulse wave velocity; RD, radial diffusivity; STROBE, Strengthening the Reporting of Observational Studies in Epidemiology; TFCE, threshold-free cluster enhancement; WAIS-IV, Wechsler Adult Intelligence Scale-Fourth Edition; WM, white matter; WML, white matter lesion.

- Shortly after that, at mean age 69 years, they underwent cognitive tests and brain magnetic resonance imaging (MRI) scans to assess the size, connections, and blood supply of different brain regions.
- We observed that a faster rate of aortic stiffening during the 4 years in mid-to-late life was linked to lower blood flow and poorer markers of brain connectivity across several brain regions.
- We also found that lower memory performance in older age was most closely linked to the first measures of aortic stiffness, taken at baseline, rather than the later measures of stiffness.

What do these findings mean?

- Faster stiffening of the aorta in mid-to-older ages may influence brain health, specifically, the brain's delicate blood vessels and the connectivity between different brain regions.
- However, in order to benefit memory in older age, strategies to prevent or slow down arterial stiffening may be required even earlier in the life span.
- Although this was a large study, it was conducted in a relatively well-educated and largely male sample, so future research on more representative samples are needed to allow us to generalise the findings to the wider population.

Introduction

Nearly 50 million people currently live with dementia worldwide, and this figure is expected to triple by 2050 [1]. While the exact mechanisms underlying dementia are still poorly understood, accumulating evidence implicates a number of potentially modifiable risk factors linked to another global health burden, cardiovascular disease (CVD) [2]. Timely prevention strategies targeting risk factors common to these diseases may therefore have a dual benefit in reducing 2 of the world's most prevalent causes of morbidity and mortality.

One established risk factor for CVD is disproportionate stiffening of the aorta [3]. Aortic stiffness refers to the loss of elasticity in the artery wall and occurs gradually with age [4]. Acceleration of this process can also compromise the artery's ability to dampen the pulsatile energy flowing from the heart to delicate target organs like the brain. Indeed, numerous studies have now demonstrated a relationship between aortic stiffness and accelerated cognitive decline, suggesting that this may also be an important risk factor for dementia ([5–8]; for review, see[9]). Less well understood, however, is the relationship of aortic stiffness with cerebrovascular and microstructural changes in the brain that may contribute to cognitive impairment and dementia [10].

Some magnetic resonance imaging (MRI) studies have revealed a link between higher aortic stiffness and markers of vascular pathology in the brain such as white matter lesions (WMLs), lacunar infarcts, and cerebral microbleeds ([11–13]; for review, see[14]). However, most of these studies have visually dichotomized brain tissue into normal and abnormal, which lacks quantitative and spatial detail about the specific brain regions that may be most sensitive to

damage. Very few studies have examined brain microstructural [15,16] or cortical perfusion [17] correlates of aortic stiffness at a voxel level.

Moreover, to our knowledge, no human study to date has investigated the potential impact that faster rates of arterial stiffening have on these measures. An increased rate of arterial stiffening is commonly observed in the transition from mid-to-late life; a time when a combination of ageing and long-term CVD risk exposure interact to negatively influence distensibility of major arteries. Recent work in animal models has strengthened claims that this link between excess pulsatility and cerebral outcomes may be causal [18]. De Montgolfier and colleagues showed that surgically induced rapid increases in pulsatile pressure in mice results in endothe-lial dysfunction and hypoperfusion within the brain's fragile microcirculation, all accompanied by significant declines in cognitive function [18].

Using repeat measures of carotid–femoral pulse wave velocity (PWV) measured as part of the long-running Whitehall II study, we assessed changes in arterial stiffness over 4 years and investigated their relationship with multimodal brain MRI and cognitive outcomes in a subset of 542 participants of the Whitehall II Imaging Sub-study.

Methods

Sample selection

Participants were drawn from the Whitehall II Study, a cohort of 10,308 British civil servants followed since 1985 [19]. Arterial stiffness was measured using carotid-femoral PWV at Phase 9 (2007 to 2009, *n* = 4,347) and again at Phase 11 (2012 to 2013, *n* = 4,344), and a total of 3,454 participants had repeat measures of PWV in the Whitehall II cohort [20]. This cohort has had a consistently high retention rate since its inception, with 87% of Phase 9 participants returning for a clinical visit at Phase 11. Following Phase 11, 774 randomly selected participants received multimodal brain MRI scans and cognitive tests at the Wellcome Centre for Integrative Neuroimaging, University of Oxford, as part of the Whitehall II Imaging Sub-study (2012 to 2016, protocol described previously[21]). Due to a scanner upgrade two-thirds of the way through the study, 2 MRI scanners were used: a 3T Siemens Magnetom Verio scanner (Erlangen, Germany) with a 32-channel head coil (n = 552, April 2012 to December 2014) and a 3T Siemens Prisma Scanner (Erlangen, Germany) with a 64-channel head-neck coil in the same centre (n = 222, July 2015 to December 2016) [21]. The scan parameters were identical or closely matched between scanners (S1 Text), and a scanner covariate was used in all analyses. Pseudocontinuous arterial spin labelling (pCASL) scans to quantify cerebral blood flow (CBF) were acquired in a subset of 145 participants who were all scanned on the Verio scanner.

Of the 774 participants, we excluded those with incidental findings on the MRI scan (e.g., large strokes, brain tumours, or cysts; n = 31), incomplete or poor quality T1 or diffusion tensor imaging (DTI) data (n = 59), incomplete PWV measurements at either Phase 9 or 11 (n = 140), and extreme outlying values of PWV at Phases 9 and 11, >4 SDs from the mean (n = 2). Accordingly, to maximise power for the voxel-wise cross-subject statistics, the volumetric and diffusion analyses were performed on 542 participants, while the WML analysis was performed in a subset of n = 533 with available FLAIR scans, cognitive analyses in a subset of n = 537 with all available cognitive data, and the perfusion analysis in a subset of n = 112 with available pCASL scans. None of the participants had dementia at the time of the MRI scan. A flowchart of the sample selection is in S1 Fig, and a comparison of the selected sample and full imaging cohort is presented in S1 Table. This study was not preregistered, but the hypothesis and analysis protocols are within the scope of planned hypotheses for this cohort, as published in the Whitehall II Imaging Study Protocol [21].

Data analysis of the vascular, cognitive, and MRI data for this study was conducted between May 2018 and January 2020. The manuscript has been modified from its original submission in accordance with peer review in August 2020, with the addition of new supplementary files (S1–S3 Tables and S2 Text) and the revision of Tables 1–3, Figs 1 and 2, and S1 and S2 Figs as a result of the exclusion of 2 participants who had outlying values of PWV at both phases. Outlier removal strengthened all the associations reported in the original manuscript but did not alter the overall conclusions. This study is reported as per the Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) guideline for cohort studies.

Ethics statement

The study was approved by the University of Oxford Medical Sciences Interdivisional Research Ethics Committee (Reference: MS IDREC-C1-2011-71) and the University College London Committee on the Ethics of Human Research (Reference: 85/0938). Written informed consent was obtained from all participants.

Pulse wave velocity

The protocol for measuring PWV in the Whitehall II Study has been described previously [6]. Briefly, aortic PWV was assessed between carotid and femoral sites using applanation tonometry (SphygmoCor; Atcor Medical, Australia). At each phase, 2 measurements of PWV were acquired from each participant, and a third measurement was taken if the difference between the first 2 measurements was >0.5 m/s. Path length was determined with a tape measure by subtracting the carotid–sternal notch distance from the femoral–sternal notch distance. Repeat measures of PWV at Phases 9 and 11 yielded a bias and 95% limits of agreement of 0.03 (–2.3, 2.4) and –0.01 (–2.6, 2.6) m/s, respectively. The average of the measurements at each phase was used in cross-sectional analyses. The change (Δ PWV) was calculated as the difference

Table 1. Characteristics of the 542 participants at Phase 9 and Phase 11.

Sample characteristics	Phase 9 2007–2009	Phase 11 2012–2013	
N	542	542	
Age, years	63.9 (5.2)	68.0 (5.2)	
Age range, years	55.6-77.4	59.8-81.8	
<i>N</i> male (%)	444/542 (81.9%)		
PWV m/s	8.1 (1.7)	8.7 (2.1)	
PWV range m/s	4.7-15.0	4.5-17.5	
Mean change in PWV between Phase 9 and Phase 11, m/s/year	0.2 (0.4)		
Systolic blood pressure, mmHg	122.5 (14.7)	125.1 (15.4)	
Diastolic blood pressure, mmHg	70.6 (10.1)	70.3 (9.5)	
MAP, mmHg	87.9 (11.0)	88.6 (10.7)	
BMI, kg/m ²	25.9 (3.6)	25.9 (3.6)	
Antihypertensive treatment, N (%)	137/542 (25.3%)	175/542 (32.3%)	
Current smokers, N (%)	29/538 (5.4%)*	20/541 (3.7%)*	
Type 2 diabetes, N (%)	8/542 (1.5%)	37/539 (6.9%)*	
Current CVD (angina, stroke, or myocardial infarction), $N(\%)$	12/542 (2.2%)	16/542 (3.0%)	

Values represent mean (SD) or no. (%).

* indicates the presence of missing data.

BMI, body mass index; CVD, cardiovascular disease; MAP, mean arterial pressure; *N*, number; PWV, pulse wave velocity; SD, standard deviation.

https://doi.org/10.1371/journal.pmed.1003467.t001

OVERVIEW $(N = 542)$	Mean (SD)				
Age at MRI phase, years	69.79 (5.22)				
Years of education	14.24 (3.01)				
Years from Phase 9 to MRI phase	5.90 (1.40)				
Years from Phase 11 to MRI phase	1.82 (1.36)				
Years from Phase 9 to Phase 11	4.08 (0.18)				
STRUCTURAL OUTCOMES (N = 542)					
GM volume, % ICV	51.5 (4.2)				
Global FA	0.48 (0.02)				
Global MD (× 10^{-3})	0.69 (0.03)				
Global RD (× 10^{-3})	0.49 (0.03)				
Global AD (× 10^{-3})	1.08 (0.03)				
WMLs, % ICV ^a	0.53 (0.33)				
CBF OUTCOMES (CBF, N = 112)					
Frontal lobe CBF (ml/100 g/min)	51.37 (12.17)				
Parietal lobe CBF (ml/100 g/min)	62.06 (15.49)				
Occipital lobe CBF (ml/100 g/min)	49.04 (18.82)				
Temporal lobe CBF (ml/100 g/min)	49.35 (10.17)				
COGNITIVE OUTCOMES (N = 537)					
Verbal learning (no. of words recalled, short term)	27.81 (4.37)				
Delayed verbal recall (no. of words recalled, delayed)	9.45 (2.51)				
Semantic fluency (no. of animals listed)	22.51 (5.28)				
Short-term memory (total digit span score)	31.08 (5.47)				
Executive function (Trail Making Test B-A, min)	0.58 (0.44)				
Executive function (digit substitution score)	63.45 (13.05)				
MoCA (total score)	27.39 (2.10)				

Table 2. Summary of the brain and cognitive outcome variables examined at the MRI phase (2012-2016).

^aAnalysed WML subset (N = 533).

Values represent mean (SD). GM volume and WMLs are expressed as % of total intracranial volume. AD, axial diffusivity; CBF, cerebral blood flow; FA, fractional anisotropy; GM, grey matter; ICV, intracranial volume; MD, mean diffusivity; MoCA, Montreal Cognitive Assessment; MRI, magnetic resonance imaging; RD, radial diffusivity; SD, standard deviation; WML, white matter lesion.

https://doi.org/10.1371/journal.pmed.1003467.t002

between phases divided by time between visits (m/s/year), with positive values representing faster rates of stiffening.

Neuroimaging outcomes

Detailed protocols for the acquisition and preprocessing of all scans have been described previously[21] and in S1 Text. All images were processed using FMRIB Software Library (FSL) tools. Briefly, grey matter (GM) volumes were assessed using voxel-based morphometry (FSL-VBM) analyses on structural T1-weighted scans [22]. WMLs were quantified as hyperintensities on fluid-attenuated inversion recovery (FLAIR) scans using the FSL-BIANCA tool [23]. CBF was quantified as the rate of delivery of arterial blood to brain tissue (ml of blood per 100 g of tissue per minute) using pCASL scans analysed with the FSL-BASIL toolkit [24]. White matter (WM) microstructure was assessed using DTI scans analysed with tract-based spatial statistics (FSL-TBSS) [25]. DTI is sensitive to directional diffusion of water within the axon, which is unrestricted along the axon, but hindered perpendicularly due to the presence of the myelin sheath. Directionality and amount of water diffusion is quantified by DTI-

|--|

	Model 1: Phase 9 PWV (m/s)		Model 2: ΔPWV (m/s/y)		Model 3: Phase 11 PWV (m/s)			
	B [95% CI]	p	B [95% CI]	p	B [95% CI]	p		
1. Brain structure (N = 542)								
$FA[\times 10^3]$	-0.90 [-1.90, 0.09]	0.08	-5.65 [-9.75, -1.54]	0.007*	-1.12 [-1.95, -0.29]	0.009*		
$MD [\times 10^6]$	1.19 [-0.19, 2.57]	0.09	5.66 [-0.06, 11.38]	0.05	1.22 [0.06, 2.38]	0.04		
RD $[\times 10^6]$	1.39 [-0.10, 2.88]	0.07	7.50 [1.36, 13.64]	0.017	1.56 [0.32, 2.81]	0.014		
$AD [\times 10^6]$	0.79 [-0.54, 2.12]	0.25	1.97 [-3.56, 7.50]	0.48	0.54 [-0.59, 1.66]	0.35		
GM (%ICV)	-0.06 [-0.27, 0.16]	0.60	-0.41 [-1.30, 0.48]	0.37	-0.09 [-0.27, 0.09]	0.32		
WML (%ICV, <i>N</i> = 533)	0.001 [-0.02, 0.02]	0.91	0.03 [-0.04, 0.10]	0.39	0.002 [-0.01, 0.02]	0.78		
2. CBF (ml/100 g/min, N = 112)								
Frontal lobe	0.59 [-1.14, 2.32]	0.50	-10.85 [-17.91, -3.79]	0.003*	-1.05 [-2.43, 0.33]	0.14		
Temporal lobe	0.61 [-0.80, 2.01]	0.39	-7.14 [-12.94, -1.35]	0.016	-0.42 [-1.55, 0.72]	0.47		
Parietal lobe	0.50 [-1.65, 2.66]	0.65	-12.75 [-21.58, -3.91]	0.005*	-1.28 [-3.00, 0.45]	0.15		
Occipital lobe	0.61 [-1.94, 3.17]	0.64	-13.30 [-23.87, -2.73]	0.014	-1.41 [-3.43, 0.62]	0.17		
3. Cognitive performance (N = 537)								
Semantic fluency	-0.47 [-0.76, -0.18]	0.001*	0.89 [-0.32, 2.11]	0.15	-0.12 [-0.37, 0.12]	0.32		
Verbal learning	-0.36 [-0.60, -0.12]	0.003*	0.44 [-0.55, 1.43]	0.38	-0.10 [-0.30, 0.10]	0.33		
Delayed verbal recall	-0.11 [-0.25, 0.03]	0.11	0.18 [-0.40, 0.77]	0.54	-0.01 [-0.13, 0.11]	0.91		
Short-term memory (digit span)	-0.03 [-0.33, 0.28]	0.86	-0.19 [-1.46, 1.08]	0.77	-0.08 [-0.34, 0.17]	0.52		
Executive function (trail making)	-0.001 [-0.03, 0.02]	0.91	-0.01 [-1.11, 0.09]	0.84	-0.002 [-0.02, 0.02]	0.85		
Executive function (digit substitution)	-0.33 [-1.04, 0.39]	0.37	1.54 [-1.42, 4.50]	0.31	-0.05 [-0.66, 0.55]	0.86		
Global cognition (MoCA)	-0.12 [-0.24, -0.01]	0.04	-0.08 [-0.57, 0.41]	0.75	-0.07 [-0.17, 0.03]	0.17		

Models 1 and 3 were adjusted for age, MAP, BMI, and antihypertensive use at the respective phase (9 or 11), years from the respective phase to the MRI scan and scanner model, sex, education, and socioeconomic grade. Model 2 was adjusted for the aforementioned covariates as well as Phase 9 PWV and years from P11 to MRI to examine the unique contribution of rate of stiffening (regardless of baseline). Unstandardised B coefficients (B) and 95% CIs are reported, representing mean change in outcome per 1 m/s increase in PWV (for Models 1 and 3) and per 1m/s/year increase in rate of change of PWV (for Model 2). GM volume and WMLs are expressed as percentages of ICV; CBF is expressed in ml/100 g tissue/min; PWV in m/s, Δ PWV in m/s/year. FA values were multiplied by 103 and RD, MD, AD values by 106. * indicates associations which survive post hoc Bonferroni corrections for multiple comparisons.

AD, axial diffusivity; BMI, body mass index; CBF, cerebral blood flow; CI, confidence interval; FA, fractional anisotropy; GM, grey matter; ICV, intracranial volume; MAP, mean arterial pressure; MD, mean diffusivity; MoCA, Montreal Cognitive Assessment; MRI, magnetic resonance imaging; PWV, pulse wave velocity; RD, radial diffusivity; WML, white matter lesions.

https://doi.org/10.1371/journal.pmed.1003467.t003

derived metrics such as fractional anisotropy (FA), radial diffusivity (RD), axial diffusivity (AD), and mean diffusivity (MD).

Decreases in FA alongside increases in diffusivity are established and sensitive measures of WM damage in dementia [26]. Lower CBF and higher WMLs are also well-established surrogate neuroimaging end points for dementia [27].

In addition to voxel-based analyses, we also examined summary measures of global brain physiology. For each participant, total GM and WML volumes were extracted using FSL-FAST and FSL-BIANCA, respectively. These were normalised to total intracranial volume and expressed as GM% and WML%. Global FA, MD, RD, and AD were extracted from the mean TBSS skeleton, with values ranging from 0 (representing isotropic diffusion) to 1 (representing anisotropic diffusion). Total CBF from the frontal, temporal, parietal, and occipital lobes were extracted from native-space CBF maps using MNI152 masks and expressed in ml/100 g/min.

Cognitive outcomes

We examined performance on 7 cognitive domains: semantic fluency: number of animals named in 1 minute; verbal learning: number of words learned after 3 trials of the Hopkins



Fig 1. Association of aortic stiffening with WM microstructure and CBF. Higher aortic stiffening was associated with (A) higher RD, (B) higher AD, (C) lower FA, (D) higher MD, and (E) lower CBF. Associations with Phase 9 PWV, Phase 11 PWV, and Δ PWV are presented in pink, blue, and red yellow, respectively. Five horizontal slices are displayed with MNI152 coordinates ranging from Z = 65 to Z = 105 for WM and Z = 56 to Z = 104 for CBF. The WM clusters are overlaid on the study-specific mean FA skeleton (green) and the standard FMRIB58 FA image, and the CBF clusters are overlaid on the standard MNI152 brain. All results are thresholded at *p* < 0.05 (TFCE and FWE-corrected *p*-values) and are presented with a colour gradient for 1 *p*-values. AD, axial diffusivity; CBF, cerebral blood flow; FA, fractional anisotropy; FWE, family-wise error; L, left; MD, mean diffusivity; PWV, pulse wave velocity; R, right; RD, radial diffusivity; TFCE, threshold-free cluster enhancement; WM, white matter.

https://doi.org/10.1371/journal.pmed.1003467.g001

Verbal Learning Test–Revised (HVLT-R); delayed verbal recall: number of words recalled after a 30-minute delay on HVLT-R; working memory: total score on the Digit Span Forward, Backward, and Sequence tests; executive function: difference in time to completion between Trail Making Test B and A and digit substitution test (Wechsler Adult Intelligence Scale-Fourth Edition (WAIS-IV)); and global cognitive function: Montreal Cognitive Assessment (MoCA).

Statistical analysis

All exposures and outcomes were treated as continuous measurements. Longitudinal change in PWV between Phase 9 and Phase 11 was analysed using an ANCOVA covarying for base-line age and sex.



Fig 2. Partial regression plots showing the associations of PWV with brain and cognitive outcomes which survived Bonferroni correction for multiple comparisons. Plots present estimated marginal means for (A, B) Predicted global FA, (C) Predicted CBF from frontal (green) and parietal (orange) lobes, and (D) Predicted memory performance on semantic fluency (green) and verbal learning (orange) tests, plotted against (A,C) Rate of change of PWV, (B) PWV at Phase 11, and (D) PWV at Phase 9. Estimated marginal means are derived from linear regression models adjusted for age, MAP, BMI, and antihypertensive treatment measured at the respective phase (9 or 11), number of years from the respective phase (9 or 11) to the MRI scan, and sex, education, socioeconomic status, and MRI scanner model. BMI, body mass index; CBF, cerebral blood flow; FA, fractional anisotropy; MAP, mean arterial pressure; MRI, magnetic resonance imaging; PWV, pulse wave velocity.

https://doi.org/10.1371/journal.pmed.1003467.g002

Cross-subject voxel-wise regressions were performed separately on spatial maps of GM volume, WM microstructure (FA, MD, RD, and AD) and CBF (preprocessing detailed in <u>S1 Text</u>). For each modality, the respective spatial maps for all participants were concatenated into a 4D file. This was submitted to the FSL-Randomise tool, which is a nonparametric permutationbased method used for inference (thresholding) on statistical maps when the null distribution is not known (<u>https://fsl.fmrib.ox.ac.uk/fsl/fslwiki/Randomise/Theory</u>, [28]). It allows the use of a standard general linear model (GLM) design matrix to model explanatory variables (in this case PWV and covariates). Here, we used 5,000 permutations in FSL-Randomise, which greatly reduces the margin of error for the nominal alpha, and examined the threshold-free cluster enhancement (TFCE)-based *t* tests, which are more sensitive and interpretable than clusterbased thresholding [29]. Three GLMs were run to assess the associations of Phase 9 PWV (Model 1), Δ PWV (Model 2), and Phase 11 PWV (Model 3) on the aforementioned MRI and cognitive outcomes. All models included the covariates listed below. For the MRI outcomes, voxel-wise results are displayed at a TFCE-corrected and family-wise error (FWE)-corrected significance threshold of *p* < 0.05, corrected for multiple voxel-wise comparisons. This represents a 95% confidence of no false positives in the significant clusters identified.

Statistical analyses of MRI-derived variables were performed in R (Version 3.5.2). Four CBF regions of interest (frontal, temporal, parietal, and occipital lobes), 4 global WM metrics (global FA, MD, RD, and AD), and 7 cognitive tests were used as dependent variables in separate multivariate GLMs to test for associations with Phase 9 PWV (Model 1), Δ PWV (Model 2), and Phase 11 (Model 3). In each case, if the multivariate model was significant, post hoc univariate regressions were interpreted to assess contributions of the individual CBF, WM, and cognitive measures. Post hoc tests were considered statistically significant at a strict Bonferroni-corrected threshold of *p* < 0.05/4 (i.e., *p* < 0.0125) to correct for multiple comparisons across the 4 WM and CBF metrics and *p* < 0.05/7 (i.e., *p* < 0.007) for the 7 cognitive tests. Statistical summaries for results which did not survive Bonferroni correction are reported for completion. Additionally, the associations of Phase 9 (Model 1), Δ PWV (Model 2), and Phase 11 PWV (Model 3) with WML% and GM% were assessed in separate univariate linear regression models. All models included the covariates listed below.

We further assessed whether associations of PWV with cognition were mediated by brain MRI markers using a mediation analysis in PROCESSv3.4 for SPSS (2012 to 2019 by Andrew F. Hayes; detailed in <u>S2 Text</u>).

Covariates

Covariates were selected a priori based on the literature. All models (presented in Table 3) included sex, total years of full-time education (self-reported at the MRI phase), socioeconomic status (defined based on the highest civil service employment grade achieved at Phase 3 in 2002 to 2004; 1 = highest grade; 2 = intermediate, and 3 = lowest grade), and MRI scanner model, as well as the following covariates measured at Phase 9 (for Models 1 and 2) or Phase 11 (for Model 3): age, mean arterial pressure (MAP), body mass index (BMI), current antihypertensive treatment (selfreport), and number of years from the respective phase (9 or 11) to the MRI scan. Model 2 (ΔPWV) also covaried for Phase 9 PWV, in order to assess the brain/cognitive correlates of rate of aortic stiffening regardless of baseline measures. In order to confirm that observed associations with ΔPWV (i.e., Model 2) were not driven by the inclusion of baseline PWV in the model, we also tested 2 additional models, all of which produced equivalent results (S3 Table): First, we tested associations of ΔPWV without including baseline PWV as a covariate, and secondly, we used the residuals from the regression of Phase 11 PWV on Phase 9 PWV in the place of Δ PWV—a technique which has the advantage of using an exposure which lies orthogonal to, and is therefore unrelated to, baseline values (S3 Table). We note that all models produced equivalent results, thus ensuring the robustness of our findings. Additional covariates such as self-reported current smoking status (yes/no), self-reported current CVD (defined as having been diagnosed with myocardial infarction, angina, or stroke at a clinical examination), self-reported history of type 2 diabetes (yes/no) were tested in a stepwise forward linear regression and as they did not significantly improve model fit they were not included in the final results described in Table 3 in order to avoid overfitting.

Results

Participant characteristics

Participant characteristics are in Tables 1 and 2; 81.9% of the participants were male, and 58.9%, 38.2%, and 3.0% of participants were categorised in socioeconomic grades 1, 2, and 3,

respectively. There was a significant increase in PWV during the 4-year follow-up from Phase 9 to 11 with a mean increase of 0.7 m/s (SD = 1.6, p < 0.0001) between phases, after correcting for baseline age and sex. On average, PWV increased by 0.2 (SD = 0.4) m/s per year (S2 Fig).

The samples in the structural (n = 542) and perfusion MRI analysis (n = 112) were representative of the parent Whitehall Imaging cohort (n = 774) in key characteristics such as age, years of education, MAP, current antihypertensive treatment, proportion of smokers, and those with current CVDs (S1 Table). However, the selected samples had a smaller proportion of females and a lower BMI than the parent cohort. We note that the absolute differences in these measures were very small (approximately 4% fewer females and 0.5 kg/m² reduction in BMI in the structural sample compared to the parent cohort); nonetheless, BMI and sex have been included as covariates in all analyses of PWV with brain and cognitive outcomes.

Association of PWV with brain microstructure

Higher Δ PWV (p_(model) = 0.003) and higher Phase 11 PWV (p_(model) = 0.004) predicted poor overall WM microstructure in covariate-adjusted multivariate GLMs with global FA, MD, RD, and AD as dependent variables. Post hoc univariate regressions revealed that this was driven by associations of aortic stiffening with lower global FA (B [95% CI] for Δ PWV: -5.65 [-9.75, -1.54] and Phase 11 PWV: -1.12 [-1.95, -0.29]), higher global MD (B [95% CI] for Δ PWV: 5.66 [-0.06, 11.38] and Phase 11 PWV: 1.22 [0.06, 2.38]), and higher global RD (B [95% CI] for Δ PWV: 7.50 [1.36, 13.64] and Phase 11 PWV: 1.56 [0.32, 2.81]); however, only the associations with global FA survived Bonferroni correction for multiple comparisons (*p* < 0.0125, Table 3 and Fig 2).

Voxel-wise analyses demonstrated that aforementioned associations of Δ PWV with FA, MD, and RD were localised to 23.9%, 11.8%, and 22.2% of WM tracts, respectively, covering the corpus callosum, fornix, anterior thalamic radiation, superior and posteriori corona radiata, superior longitudinal fasciculus, and corticospinal tracts (Table 4). Similar widespread associations were observed for Phase 11 PWV, covering 22.3%, 13.6%, and 24.1% of WM for FA, MD, and RD, respectively (Table 4). For both Δ PWV and Phase 11 PWV, voxel-wise analyses revealed additional, localised, and less pronounced associations with AD within the uncinate fasciculus and posterior corona radiata (Fig 1 and Table 4). Furthermore, while evidence of associations were also seen with baseline Phase 9 PWV; (p_(model) = 0.18), Table 3] and voxel-based analyses (Fig 2, covering only 1.4% of WM tracts for RD).

Neither cross-sectional nor longitudinal aortic PWV were associated with WML%, total GM% (Table 3), or GM volume assessed with voxel-wise analysis.

Association of PWV with cerebral perfusion

Higher Δ PWV was associated with lower total CBF in the frontal (B [95% CI]: -10.85 [-17.91, -3.79]), temporal (B [95% CI]: -7.14 [-12.94, -1.35]), parietal (B [95% CI]: -12.75 [-21.58, -3.91]), and occipital lobes (B [95% CI]: -13.30 [-23.87, -2.73]) in the subset of 112 participants with perfusion imaging scans; however, only associations with frontal and parietal lobe CBF survived Bonferroni correction for multiple comparisons (*p* < 0.0125, Table 3 and Fig 2). The voxel-wise analyses revealed most pronounced effects in the parietal, lateral occipital, and cuneal cortex, occipital pole, right middle temporal gyrus, frontal pole, and the precuneus (Fig 1). Cross-sectional PWV at Phases 9 and 11 were not significantly associated with regional CBF in the voxel-wise analysis or using extracted mean lobar CBF (Table 3).

Outcome	Cluster no.	Number of voxels (% of WM tract)	<i>p</i> -value	Coordinates of maxima (X, Y, Z)	Location of maxima		
1. Association with ΔPWV							
FA	1	37,544/159,400 (23.6%)	0.004	73, 103, 105	Body of corpus callosum		
FA	2	45/159,400 (0.03%)	0.049	75, 83, 40	Right corticospinal tract (middle cerebellar peduncle)		
MD	1	18,853/159,400 (11.8%)	0.019	66, 95, 100	Right posterior corona radiata		
RD	1	35,406/159,400 (22.2%)	0.01	73, 106, 105	Body of corpus callosum		
AD	1	144/159,400 (0.09%)	0.046	67, 97, 101	Right posterior corona radiata		
CBF	1	37,835/206,802 (18.3%)	0.002	46, 95, 41	Frontal pole		
CBF	2	127/206,802 (0.06%)	0.031	18, 61, 25	Right middle temporal gyrus		
CBF	3	34/206,802 (0.02%)	0.044	20, 51, 51	Right parietal operculum		
2. Associat	tion with Phas	e 11 PWV					
FA	1	34,854/159,400 (21.9%)	0.01	123, 63, 73	Left posterior thalamic radiation		
FA	2	683/159,400 (0.4%)	0.047	100, 96, 136	Corticospinal tract		
MD	1	21,745/159,400 (13.6%)	0.016	115, 115, 102	Left superior corona radiata		
RD	1	38,470/159,400 (24.1%)	0.01	125, 124, 54	Left uncinate fasciculus		
AD	1	269/159,400 (0.2%)	0.046	110, 125, 105	Left superior corona radiata		
AD	2	209/159,400 (0.1%)	0.046	113, 89, 103	Left posterior corona radiata		
AD	3	177/159,400 (0.1%)	0.045	115, 133, 105	Left superior corona radiata		
3. Association with Phase 9 PWV							
RD	1	1,962/159,400 (1.2%)	0.041	123, 71, 99	Left superior longitudinal fasciculus		
RD	2	272/159,400 (0.2%)	0.049	114, 93, 123	Corticospinal tract		

Table 4. Cluster report for the voxel-wise associations of PWV with FA, MD, RD, AD, and CBF.

A total of 159,400 and 206,802 voxels were investigated for the WM tracts and CBF analyses, respectively. Cluster sizes (number of significant voxels and % of WM or GM area covered), TFCE and FWE-corrected *p*-values, voxel coordinates in MNI152 space, and locations of cluster maxima are reported for all significant clusters. AD, axial diffusivity; CBF, cerebral blood flow; FA, fractional anisotropy; FWE, family-wise error; GM, grey matter; MD, mean diffusivity; PWV, pulse wave velocity; RD, radial diffusivity; TFCE, threshold-free cluster enhancement; WM, white matter.

https://doi.org/10.1371/journal.pmed.1003467.t004

Association of PWV with cognitive performance

Baseline PWV at Phase 9 was associated with lower overall cognitive performance at the MRI phase ($p_{(model)} = 0.01$) in covariate-adjusted multivariate GLMs. Post hoc linear regressions revealed that this association was driven by poorer performance on the semantic fluency and verbal learning tests, which survived a Bonferroni-corrected threshold for significance (p < 0.007, Table 3 and Fig 2). There was also some evidence for an association with lower MoCA scores; however, this did not survive Bonferroni corrections (p < 0.05, Table 3). In contrast, neither Phase 11 PWV ($p_{(model)} = 0.71$) nor Δ PWV ($p_{(model)} = 0.76$) were associated with cognitive performance at the MRI visit, and while the post hoc univariate regressions are presented for completion, these are not further interpreted (Table 3). Moreover, the associations of Phase 9 PWV with cognitive outcomes were not mediated by WM microstructure (S2 Table).

Discussion

In this study, we show that an increased rate of arterial stiffening is associated with lower WM microstructural integrity and CBF in older age. Furthermore, these associations were present in diffuse brain areas, suggesting that exposure to excess pulsatility may result in a widespread damaging effect on the fragile cerebral microstructure. Cognitive function at follow-up related more closely with baseline arterial stiffness rather than rate of arterial stiffening. Taken together, these findings suggest that although faster rates of arterial stiffening in the transition

to old age may negatively impact brain structure and function, long-term exposure to higher levels of arterial stiffness prior to this point may be the most important determinant for future cognitive ability.

While aortic stiffening has predominantly been studied in the context of CVD, recent evidence suggests that large artery dysfunction may also play a role in dementia [30]. Indeed, patients with Alzheimer's disease and vascular dementia reportedly have higher levels of aortic stiffness relative to cognitively healthy adults [30]. Aortic stiffening is a hallmark of vascular ageing and may lead to a heightened state of oxidative and inflammatory damage within the cerebral tissues due to an increased penetrance of excess pulsatility into the fragile microcirculation of the brain.[9] These changes have been shown to disrupt endothelial cell function and the blood brain barrier in animal models and have also been hypothesised to compromise cerebral perfusion and ultimately lead to amyloid deposition, neurodegeneration, and cognitive impairment.[9,31] While previous studies have related cross-sectional measures of arterial stiffness to cognition, this is the first study, to our knowledge, to publish associations between progressive increases in aortic stiffening over a 4-year period and cerebral and cognitive outcomes in later life.

We report a number of novel findings. First, faster rates of arterial stiffening from average age 64 to 68 years, and higher follow-up aortic stiffness at approximately 68 years were associated with widespread WM microstructural decline within the corpus callosum, corona radiata, superior longitudinal fasciculus, cortico-spinal tracts, and internal capsule. Observations of lower FA with a concomitantly higher diffusivity within these tracts are suggestive of axonal degeneration and loss of myelin. Notably, these tracts have also been shown to be vulnerable to arterial stiffness in 2 smaller studies in older adults [15,16]. These tracts are supplied by the anterior and middle cerebral arteries and given that they are also compromised in Alzheimer's and vascular dementia [26], they may be potential markers of the role of arterial stiffening in dementia pathophysiology. However, in our study, associations of arterial stiffness at the younger baseline age of 64 with WM were weaker, more localised, and only restricted to voxel-wise increases in RD compared to the more widespread WM impairments linked to the later measurements of PWV. This is somewhat in contrast to findings from the Framingham Heart Study, which report associations of PWV with lower FA in the corpus callosum and corona radiata even in younger adults (approximately 46 years old) [16]. They also note, however, that their observed associations with FA were accentuated by age. Cohort biases may play a role in reconciling these inconsistencies; however, it is also plausible that while regional (and perhaps subthreshold) PWV-related changes in cerebral microstructure can begin in early-to-mid life, these changes become more widespread and pronounced as rates of aortic stiffening increase in the transition from mid-to-late life.

Second, in addition to evidence of disruptions to WM structural integrity, we also noted significantly lower CBF in posterior cingulate, parietal, medial temporal, frontal, and occipital cortices with faster rates of aortic stiffening in a subset of 112 participants. These results are in line with a recent cross-sectional study in individuals with mild cognitive impairment [17] and implicate brain areas typically affected by vascular ageing [32]. Similar to WM metrics, these effects were found to be 5 to 10 times larger for rates of stiffening than cross-sectional PWV, suggesting that exposure to rapid increases in pulsatile stress may be particularly damaging to cerebral perfusion compared with more gradual changes over time.

Third, we observed no associations of aortic stiffness with WMLs and GM density. Previous studies which have measured artery stiffness using brachial–ankle PWV or augmentation index have reported associations with WML load in community-dwelling adults [33] and in hypertensive populations [11]. Self-selection bias arising from patients volunteering to attend this imaging sub-study means that it is possible that our cohort—which compared to the

general UK population was already relatively high functioning and with a healthier vascular profile [19]—had a relatively lower WML burden. Thus, changes in WM diffusion metrics (which have been noted to precede larger volumetric and pathological lesion changes) may have been more sensitive to the underlying structural tissue alterations at a voxel level than macroscopic WMLs. Moreover, previous voxel-based analyses of GM density have yielded mixed results; the Reykjavik Study reported no relationship between elevated aortic stiffness and GM in older adults [12], whereas the Framingham Heart Study revealed an association with lower thalamic volume in middle-aged adults [34]. Future studies examining longitudinal changes in GM atrophy or WML burden would be more informative than measures from a single time point.

Fourth, we observed an inverse association between baseline PWV and semantic fluency and verbal learning outcomes-alongside some evidence of an association with poor performance on the MoCA—a commonly used test for amnestic mild cognitive impairment [35]. These associations were not mediated by cerebral microstructure and are in agreement with numerous other studies linking cross-sectional measures of arterial stiffness in mid-life to later cognitive function [6], including studies in which similar relationships to verbal learning [7], episodic memory [8], and amnestic mild cognitive impairment have already been reported [36]. In contrast to our observed relationships between arterial stiffness and cerebral structure and perfusion, the cognitive outcomes were not related to later-life arterial stiffening but instead to baseline PWV alone. It is worth noting that previous reports from the Rotterdam Study and the Sydney Memory and Aging Study, which are closer in age to our Phase 11 sample (mean age approximately 68), have observed no independent associations of PWV with a similarly detailed cognitive battery examining attention, memory, and executive function in older adults [37,38], whereas the Framingham Heart Study of young adults (approximately 46 years) has reported associations with poorer processing speed and executive function [39]. Together with their findings, our results support the growing understanding that it is longterm cumulative exposure to both social (e.g., education and socioeconomic status) and physiological (inflammation, metabolic, vascular, and arterial stiffening) risk factors in the years preceding middle age-rather than the contemporary later-life changes-that are likely to represent the most important risk factor for overall cognitive ability in old age [1]. The fact that we observed considerably weaker associations between follow-up PWV and future cognitive outcomes suggests that earlier measures of PWV taken before the potential occurrence of rapid increases in arterial stiffness may represent a more effective biomarker for cognitive dysfunction risk stratification. Further studies aimed at untangling the cross-sectional and longitudinal associations of aortic stiffness with cognitive and cerebral phenotypes are required to confirm this.

Our findings suggest 2 things. First—and in agreement with previous studies linking modifiable risk factors to later adverse outcomes—early prevention strategies to reduce life-term exposure to risk factors may be required to in order to offer maximal benefits to later-life cognition, particularly in relation to domains such as semantic fluency and verbal learning. Second, novel to this study is our observation of additional relationships between faster rates of arterial stiffening during this period of life and the presence of pathological differences in WM structure and cerebral perfusion observed in the following years. These findings may provide the first evidence of the potential cerebral mechanisms underlying an accelerated rate of cognitive decline (as opposed to cross-sectional cognitive ability) during this time frame, which has previously been reported in individuals with faster rates of stiffening in the Whitehall II [40] and other cohorts [41]. We show for the first time that interventions to reduce or prevent the rapid increases in arterial pulsatility in mid-to-late life may reduce detrimental changes in WM integrity and blood flow, which have previously been linked to cognitive function, and may therefore also offer additional (albeit possibly more modest) benefits to cognitive ability in older age.

Strengths and limitations

This is 1 of the largest MRI cohorts with multimodal brain imaging phenotyping, extensive cognitive testing, and longitudinal arterial stiffness measures. Moreover, arterial stiffness was measured 6 years prior to the MRI scan, starting at a mean age of 64 years when vascular risk has been demonstrated to contribute to dementia pathology. Importantly, we report associations of PWV over and above those of traditional vascular risk factors such as MAP, BMI, and antihypertensive use, suggesting that our results reflect true intrinsic effects of aortic stiffening, thus strengthening its role as a biomarker of brain and cognitive decline.

However, as with all observational cohort studies, it was not possible to infer causal associations between these exposures and outcomes. We were also unable to examine the role of the apolipoprotein E4 (APOE4) allele on our observed associations, given the substantial proportion of participants (nearly 25% of cohort) with missing APOE information. Future studies should consider investigating this allele given its established risk for CVD and dementia [42]. To account for the extensive cerebral phenotyping across measures of microstructure and perfusion, we performed strict corrections for multiple comparisons, and results that do not survive these corrections must therefore be interpreted with caution. Moreover, the generalizability of our findings may be limited by the fact that while the Whitehall II Study comprises community-dwelling adults, it might not be entirely representative of the UK population. First, the Whitehall II cohort is predominantly male (reflecting the makeup of the British Civil Service in 1985). It is well established in the literature that women show greater increases in arterial pulse pressure from midlife onwards compared to men and that this may occur through different underlying mechanisms to those captured using carotid-femoral PWV as employed here [43]. As such, care should be taken when generalising our findings to both sexes, and future reproduction studies in gender balanced cohorts will be needed to clarify our results. Second, given the relatively high educational attainment in this cohort, it possible that we have underestimated the observed association between PWV and cognitive outcomes. Third, while the sample analysed in this study was largely representative of the parent Whitehall Imaging cohort in key demographics and cardiovascular measures, it had a higher proportion of males and lower BMI than the original cohort. However, as we have adjusted for sex, BMI, and education in all our regression models, it is unlikely that these variables have significantly driven our results. Fourth, while we observed widespread and robust associations of aortic stiffening with cerebral hypoperfusion, we note that this analysis was only conducted in a small subset (20%) of the full sample. Thus, our findings must be replicated in independent studies before generalising to the wider population.

Conclusions

To our knowledge, we have provided the first evidence of a relationship between faster rates of aortic stiffening during a 4-year period in mid-to-late life and detrimental changes to cerebral WM microstructural integrity and perfusion. These preliminary findings suggest that diffusion and perfusion MRI may be sensitive markers of recent damage caused by aortic stiffening and could therefore be potentially relevant outcome measures in intervention studies. However, as measures of cognition were found to most closely relate to baseline measures of stiffness, such interventions may need to be targeted earlier in life in order to offer maximal benefit for both vascular and cognitive health.

Supporting information

S1 Fig. Flowchart for sample selection. (TIF)

S2 Fig. Distribution of the rate of change of aortic stiffening for 542 participants. Mean change was 0.2 ± 0.4 m/s per year, ranging from -1.2 to 1.5. (TIF)

S1 Text. MRI image acquisition and MRI image analysis. MRI, magnetic resonance imaging.

(DOCX)

S2 Text. Mediation analysis. (DOCX)

S1 Table. Sample characteristics at the MRI phase (2012–2016) for the full Whitehall II Imaging cohort (N = 774) and the subsamples selected for the structural (N = 542) and perfusion (N = 112) MRI analyses. The population means or frequencies from the full cohort were compared with observed sample means or frequencies using 1-sample *t* tests (for means) or chi-squared tests (for frequencies), and the corresponding *p*-values are reported below. MRI, magnetic resonance imaging. (DOCX)

S2 Table. Results of 95% CIs of mediation (5,000× Bootstrapping). CIs excluding 0 are highlighted in bold. CI, confidence interval; FA, fractional anisotropy; M, mediator variable; PWV, pulse wave velocity; RD, radial diffusivity; SF, semantic fluency; VL, verbal learning; X, predictor variable; Y, outcome variable. (DOCX)

S3 Table. Associations of rate of arterial stiffening (Δ PWV) with brain and cognitive outcomes, with and without adjustment for baseline PWV. Model 1: full model as presented in Table 3, adjusting for sex, education, socioeconomic grade, scanner model, baseline age, BMI, MAP, antihypertensive medication, years from Phase 9 to MRI, years from Phase 11 to MRI, and baseline PWV. Model 2 was the same as Model 1 but excluded baseline PWV. Model 3 tested the association with the residuals from the regression of Phase 11 PWV on Phase 9 PWV instead of associations with Δ PWV. Thus, Model 3 did not include Δ PWV and baseline PWV as covariates; however, all the other covariates remained the same. The models show that the overall associations of rate of arterial stiffening with brain/cognition remain similar with and without adjustment for baseline PWV. AD, axial diffusivity; BMI, body mass index; CBF, cerebral blood flow; CI, confidence interval; FA, fractional anisotropy; GM, grey matter, ICV, intracranial volume; MAP, mean arterial pressure; MD, mean diffusivity; MoCA, Montreal Cognitive Assessment; MRI, magnetic resonance imaging; PWV, pulse wave velocity; RD, radial diffusivity, WML, white matter lesion. (DOCX)

Acknowledgments

We thank all the participating civil service departments; the British Occupational Health and Safety Agency; the British Council of Civil Service Unions; all participating civil servants in the Whitehall II Study; and all members of the Whitehall II Study team at University College London who so helpfully collaborated with us. The Whitehall II Study team comprises research scientists, statisticians, study coordinators, nurses, data managers, administrative assistants, and data entry staff who make the study possible. We also thank the staff at the Wellcome Centre for Integrative Neuroimaging in Oxford, in particular research radiographers Michael Sanders, MSc, Jon Campbell, MMRTech, BcAppSc, Caroline Young, DCR(R), and David Parker, BSc(Hons), who acquired the scans. Martin R. Turner, MA, MBBS, PhD, FRCP (Wellcome Centre for Integrative Neuroimaging, Oxford, United Kingdom) and his colleagues advised on incidental findings and taking over clinical responsibility for such participants. We are grateful to Professor Michael A. Chappell (DPhil) and Dr Thomas W. Okell (DPhil) for their advice on the FSL-BASIL tool for perfusion MRI analysis. No compensation was provided for staff contributions to this study.

The views expressed are those of the authors and not necessarily those of the NHS, the NIHR, or the Department of Health.

Author Contributions

Conceptualization: Sana Suri, Scott T. Chiesa, John E. Deanfield, Klaus P. Ebmeier.

Data curation: Sana Suri, Enikő Zsoldos, Nicola Filippini, Abda Mahmood, Eric J. Brunner.

- Formal analysis: Sana Suri, Scott T. Chiesa, Nicola Filippini, Ludovica Griffanti.
- **Funding acquisition:** Clare E. Mackay, Archana Singh-Manoux, Mika Kivimäki, John E. Deanfield, Klaus P. Ebmeier.
- Investigation: Sana Suri, Scott T. Chiesa, Enikő Zsoldos, Nicola Filippini, Abda Mahmood, Martin J. Shipley, Eric J. Brunner.
- Methodology: Sana Suri, Scott T. Chiesa, Martin J. Shipley, Eric J. Brunner, Klaus P. Ebmeier.
- Project administration: Enikő Zsoldos, Nicola Filippini, Abda Mahmood, Archana Singh-Manoux, Mika Kivimäki, Klaus P. Ebmeier.
- Supervision: Clare E. Mackay, Archana Singh-Manoux, Mika Kivimäki, John E. Deanfield, Klaus P. Ebmeier.
- Writing original draft: Sana Suri, Scott T. Chiesa.
- Writing review & editing: Sana Suri, Scott T. Chiesa, Enikő Zsoldos, Clare E. Mackay, Nicola Filippini, Ludovica Griffanti, Abda Mahmood, Archana Singh-Manoux, Martin J. Shipley, Eric J. Brunner, Mika Kivimäki, John E. Deanfield, Klaus P. Ebmeier.

References

- Livingston G, Huntley J, Sommerlad A, Ames D, Ballard C, Banerjee S, et al. Dementia prevention, intervention, and care: 2020 report of the Lancet Commission. Lancet. 2020. https://doi.org/10.1016/ S0140-6736(20)30367-6 PMID: 32738937
- 2. WHO. Global status report on noncommunicable diseases 2010. [cited 27 Oct 2020]. Available from: https://www.who.int/nmh/publications/ncd_report2010/en/.
- Mitchell GF, Hwang S-J, Vasan RS, Larson MG, Pencina MJ, Hamburg NM, et al. Arterial stiffness and cardiovascular events: the Framingham Heart Study. Circulation. 2010; 121:505–11. <u>https://doi.org/10. 1161/CIRCULATIONAHA.109.886655</u> PMID: 20083680
- Brunner EJ, Shipley MJ, Witte DR, Singh-Manoux A, Britton AR, Tabak AG, et al. Arterial stiffness, physical function, and functional limitation: The whitehall II study. Hypertension. 2011. https://doi.org/ 10.1161/HYPERTENSIONAHA.110.168864 PMID: 21444833
- Meyer ML, Palta P, Tanaka H, Deal JA, Wright J, Knopman DS, et al. Association of Central Arterial Stiffness and Pressure Pulsatility with Mild Cognitive Impairment and Dementia: The Atherosclerosis Risk in Communities Study-Neurocognitive Study (ARIC-NCS). J Alzheimers Dis. 2017. <u>https://doi.org/ 10.3233/JAD-161041</u> PMID: 28222517

- Araghi M, Shipley MJ, Wilkinson IB, McEniery CM, Valencia-Hernández CA, Kivimaki M, et al. Association of aortic stiffness with cognitive decline: Whitehall II longitudinal cohort study. Eur J Epidemiol. 2019. https://doi.org/10.1007/s10654-019-00586-3 PMID: 31776832
- Waldstein SR, Rice SC, Thayer JF, Najjar SS, Scuteri A, Zonderman AB. Pulse pressure and pulse wave velocity are related to cognitive decline in the Baltimore longitudinal study of aging. Hypertension. 2008. https://doi.org/10.1161/HYPERTENSIONAHA.107.093674 PMID: 18025297
- Elias MF, Robbins MA, Budge MM, Abhayaratna WP, Dore GA, Elias PK. Arterial pulse wave velocity and cognition with advancing age. Hypertension. 2009. https://doi.org/10.1161/HYPERTENSIONAHA. 108.126342 PMID: 19237680
- Iulita MF, Noriega de la Colina A, Girouard H. Arterial stiffness, cognitive impairment and dementia: confounding factor or real risk? J Neurochem. 2018; 144:527–48. https://doi.org/10.1111/jnc.14235 PMID: 28991365
- Badji A, Sabra D, Bherer L, Cohen-Adad J, Girouard H, Gauthier CJ. Arterial stiffness and brain integrity: A review of MRI findings. Ageing Res Rev. 2019. https://doi.org/10.1016/j.arr.2019.05.001 PMID: 31063866
- Henskens LHG, Kroon AA, Van Oostenbrugge RJ, Gronenschild EHBM, Fuss-Lejeune MMJJ, Hofman PAM, et al. Increased aortic pulse wave velocity is associated with silent cerebral small-vessel disease in hypertensive patients. Hypertension. 2008. <u>https://doi.org/10.1161/HYPERTENSIONAHA.108</u>. 119024 PMID: 18852384
- Mitchell GF, Van Buchem MA, Sigurdsson S, Gotal JD, Jonsdottir MK, Kjartansson L, et al. Arterial stiffness, pressure and flow pulsatility and brain structure and function: The Age. Gene/Environment Susceptibility-Reykjavik Study Brain. 2011. https://doi.org/10.1093/brain/awr253 PMID: 22075523
- van Sloten TT, Mitchell GF, Sigurdsson S, van Buchem MA, Jonsson PV, Garcia ME, et al. Associations between arterial stiffness, depressive symptoms and cerebral small vessel disease: Cross-sectional findings from the AGES-Reykjavik study. J Psychiatry Neurosci. 2016. <u>https://doi.org/10.1503/jpn.</u> 140334 PMID: 26505140
- Singer J, Trollor JN, Baune BT, Sachdev PS, Smith E. Arterial stiffness, the brain and cognition: A systematic review. Ageing Res Rev. 2014. https://doi.org/10.1016/j.arr.2014.02.002 PMID: 24548924
- Tarumi T, de Jong DLK, Zhu DC, Tseng BY, Liu J, Hill C, et al. Central artery stiffness, baroreflex sensitivity, and brain white matter neuronal fiber integrity in older adults. Neuroimage. 2015; 110:162–70. https://doi.org/10.1016/j.neuroimage.2015.01.041 PMID: 25623500
- Badji A, Noriega de la Colina A, Karakuzu A, Duval T, Desjardins-Crépeau L, Joubert S, et al. Arterial stiffness and white matter integrity in the elderly: A diffusion tensor and magnetization transfer imaging study. Neuroimage. 2019; 186:577–85. <u>https://doi.org/10.1016/j.neuroimage.2018.11.015</u> PMID: 30448213
- Jefferson AL, Cambronero FE, Liu D, Moore EE, Neal JE, Terry JG, et al. Higher Aortic Stiffness Is Related to Lower Cerebral Blood Flow and Preserved Cerebrovascular Reactivity in Older Adults. Circulation. 2018; 138:1951–62. <u>https://doi.org/10.1161/CIRCULATIONAHA.118.032410</u> PMID: 30018169
- de Montgolfier O, Pinçon A, Pouliot P, Gillis MA, Bishop J, Sled JG, et al. High Systolic Blood Pressure Induces Cerebral Microvascular Endothelial Dysfunction, Neurovascular Unit Damage, and Cognitive Decline in Mice. Hypertens (Dallas, Tex 1979). 2019. https://doi.org/10.1161/HYPERTENSIONAHA. 118.12048 PMID: 30571552
- Marmot M, Brunner E. Cohort profile: The Whitehall II study. Int J Epidemiol. 2005. <u>https://doi.org/10.1093/ije/dyh372 PMID: 15576467</u>
- Brunner EJ, Shipley MJ, Ahmadi-Abhari S, Tabak AG, McEniery CM, Wilkinson IB, et al. Adiposity, obesity, and arterial aging: longitudinal study of aortic stiffness in the Whitehall II cohort. Hypertens (Dallas, Tex 1979). 2015; 66:294–300. https://doi.org/10.1161/HYPERTENSIONAHA.115.05494 PMID: 26056335
- Filippini N, Zsoldos E, Haapakoski R, Sexton CE, Mahmood A, Allan CL, et al. Study protocol: The Whitehall II imaging sub-study. BMC Psychiatry. 2014; 14:159. https://doi.org/10.1186/1471-244X-14-159 PMID: 24885374
- Douaud G, Smith S, Jenkinson M, Behrens T, Johansen-Berg H, Vickers J, et al. Anatomically related grey and white matter abnormalities in adolescent-onset schizophrenia. Brain. 2007; 130:2375–86. https://doi.org/10.1093/brain/awm184 PMID: 17698497
- Griffanti L, Jenkinson M, Suri S, Zsoldos E, Mahmood A, Filippini N, et al. Classification and characterization of periventricular and deep white matter hyperintensities on MRI: A study in older adults. Neuroimage. 2017. https://doi.org/10.1016/j.neuroimage.2017.03.024 PMID: 28315460
- Chappell MA, Groves AR, Whitcher B, Woolrich MW. Variational Bayesian inference for a nonlinear forward model. IEEE Trans Signal Process. 2009. https://doi.org/10.1109/TSP.2009.2021636 PMID: 20221319

- Smith SM, Jenkinson M, Johansen-Berg H, Rueckert D, Nichols TE, Mackay CE, et al. Tract-based spatial statistics: voxelwise analysis of multi-subject diffusion data. Neuroimage. 2006; 31:1487–505. https://doi.org/10.1016/j.neuroimage.2006.02.024 PMID: 16624579
- Suri S, Topiwala A, Mackay CE, Ebmeier KP, Filippini N. Using Structural and Diffusion Magnetic Resonance Imaging To Differentiate the Dementias. Curr Neurol Neurosci Rep. 2014. <u>https://doi.org/10.1007/s11910-014-0475-3 PMID: 25030502</u>
- Iturria-Medina Y, Sotero RC, Toussaint PJ, Mateos-Pérez JM, Evans AC, Weiner MW, et al. Early role of vascular dysregulation on late-onset Alzheimer's disease based on multifactorial data-driven analysis. Nat Commun. 2016; 7:11934. https://doi.org/10.1038/ncomms11934 PMID: 27327500
- Winkler AM, Ridgway GR, Webster MA, Smith SM, Nichols TE. Permutation inference for the general linear model. Neuroimage. 2014. https://doi.org/10.1016/j.neuroimage.2014.01.060 PMID: 24530839
- Smith SM, Nichols TE. Threshold-free cluster enhancement: Addressing problems of smoothing, threshold dependence and localisation in cluster inference. Neuroimage. 2009. <u>https://doi.org/10.1016/j.neuroimage.2008.03.061</u> PMID: 18501637
- Hanon O, Haulon S, Lenoir H, Seux ML, Rigaud AS, Safar M, et al. Relationship between arterial stiffness and cognitive function in elderly subjects with complaints of memory loss. Stroke. 2005. <u>https://</u> doi.org/10.1161/01.STR.0000181771.82518.1c PMID: 16151027
- Hughes TM, Kuller LH, Barinas-Mitchell EJM, Mackey RH, McDade EM, Klunk WE, et al. Pulse wave velocity is associated with β-amyloid deposition in the brains of very elderly adults. Neurology. 2013. https://doi.org/10.1212/01.wnl.0000435301.64776.37 PMID: 24132374
- Suri S, Topiwala A, Chappell MA, Okell TW, Zsoldos E, Singh-Manoux A, et al. Association of Midlife Cardiovascular Risk Profiles with Cerebral Perfusion at Older Ages. JAMA Netw Open. 2019. https:// doi.org/10.1001/jamanetworkopen.2019.5776 PMID: 31225888
- Ohmine T, Miwa Y, Yao H, Yuzuriha T, Takashima Y, Uchino A, et al. Association between arterial stiffness and cerebral white matter lesions in community-dwelling elderly subjects. Hypertens Res. 2008. https://doi.org/10.1291/hypres.31.75 PMID: 18360021
- Maillard P, Mitchell GF, Himali JJ, Beiser A, Tsao CW, Pase MP, et al. Effects of arterial stiffness on brain integrity in young adults from the framingham heart study. Stroke. 2016; 47:1030–6. <u>https://doi.org/10.1161/STROKEAHA.116.012949 PMID: 26965846</u>
- Nasreddine ZS, Phillips NA, Bédirian V, Charbonneau S, Whitehead V, Collin I, et al. The Montreal Cognitive Assessment, MoCA: A brief screening tool for mild cognitive impairment. J Am Geriatr Soc. 2005. https://doi.org/10.1111/j.1532-5415.2005.53221.x PMID: 15817019
- 36. Fujiwara Y, Chaves PHM, Takahashi R, Amano H, Yoshida H, Kumagai S, et al. Arterial Pulse Wave Velocity as a Marker of Poor Cognitive Function in an Elderly Community-Dwelling Population. Journals Gerontol Ser A Biol Sci Med Sci. 2005; 60:607–12.
- Poels MMF, van Oijen M, Mattace-Raso FUS, Hofman A, Koudstaal PJ, Witteman JCM, et al. Arterial stiffness, cognitive decline, and risk of dementia: the Rotterdam study. Stroke. 2007; 38:888–92. https://doi.org/10.1161/01.STR.0000257998.33768.87 PMID: 17272780
- Singer J, Trollor JN, Crawford J, O'Rourke MF, Baune BT, Brodaty H, et al. The Association between Pulse Wave Velocity and Cognitive Function: The Sydney Memory and Ageing Study. PLoS One. 2013. https://doi.org/10.1371/journal.pone.0061855 PMID: 23637918
- Pase MP, Himali JJ, Mitchell GF, Beiser A, Maillard P, Tsao C, et al. Association of Aortic Stiffness With Cognition and Brain Aging in Young and Middle-Aged Adults. Hypertension. 2016. <u>https://doi.org/10. 1161/HYPERTENSIONAHA.115.06610</u> PMID: 26754644
- 40. Chiesa S, Masi S, Shipley M, Ellins E, Halcox J, Fraser A, et al. Abstract 11343: Accelerated Aortic Stiffening Relative to Carotid Stiffness is Associated With Cognitive Decline: The Whitehall II Study | Circulation. Circulation. 2019; 140:A11343.
- Tsao CW, Himali JJ, Beiser AS, Larson MG, Decarli C, Vasan RS, et al. Association of arterial stiffness with progression of subclinical brain and cognitive disease. Neurology. 2016. <u>https://doi.org/10.1212/</u> WNL.00000000002368 PMID: 26791155
- **42.** Haan MN, Shemanski L, Jagust WJ, Manolio TA, Kuller L. The role of APOE epsilon 4 in modulating effects of other risk factors for cognitive decline in elderly persons. JAMA-JOURNAL Am Med Assoc. 1999; 282:40–6. https://doi.org/10.1001/jama.282.1.40 PMID: 10404910
- Mitchell GF, Gudnason V, Launer LJ, Aspelund T, Harris TB. Hemodynamics of increased pulse pressure in older women in the community-based age, gene/environment susceptibility-Reykjavik study. Hypertension. 2008; 51:1123–8. https://doi.org/10.1161/HYPERTENSIONAHA.107.108175 PMID: 18259005